Author's response to reviews

Title: Saving and Empowering Young Lives in Europe (SEYLE): a randomized controlled trial

Authors:

Danuta Wasserman (danuta.wasserman@ki.se)
Vladimir Carli (vladimir.carli@ki.se)
Camilla Wasserman (wassermc@childpsych.columbia.edu)
Alan Apter (asapter@gmail.com)
Judit Balazs (judit.agnes.balazs@gmail.com)
Julio Bobes (bobes@uniovi.es)
Renata Bracale (bracale@unimol.it)
Romuald Brunner (Romuald.Brunner@med.uni-heidelberg.de)
Cendrine Bursztein-Lipsicas (b_cendrine@bezeqint.net)
Paul Corcoran (paul.nsrf@iol.ie)
Doina Cosman (doina_octavianscosman@hotmail.com)
Tony Durkee (tony.durkee@ki.se)
Dana Feldman (dana.f.man@gmail.com)
Julio Gadoros (gadoros@vadasnet.hu)
Francis Guillemin (francis.guillemin@chu-nancy.fr)
Christian Haring (christian.haring@tilak.at)
Jean-Pierre Kahn (jp.kahn@chu-nancy.fr)
Michael Kaess (michael.kaess@med.uni-heidelberg.de)
Helen Keeley (mandhhealy@eircom.net)
Dragan Marusic (dragan.marusic@upr.si)
Bogdan Nemes (nemes_bogdan@yahoo.com)
Vita Postuvan (vita.postuvan@gmail.com)
Stella Reiter-Theil (s.reiter-theil@unibas.ch)
Franz Resch (Franz.Resch@med.uni-heidelberg.de)
Pilar Sáiz (frank@uniovi.es)
Marco Sarchiapone (marco.sarchiapone@gmail.com)
Merike Sisask (merike.sisask@neti.ee)
Airi Värnik (airiv@online.ee)
Christina Hoven (hoven@childpsych.columbia.edu)

Version: 2 Date: 4 February 2010

Author's response to reviews:

Dear Editors,

I would like to submit the enclosed study protocol manuscript entitled: Saving and Empowering Young Lives in Europe (SEYLE): a randomized controlled trial to the BMC Public Health Journal.

SEYLE is funded by EU and is a mental health promoting project targeting adolescents in European schools. The study comprises 11,000 adolescents
emitted from randomized schools in 11 European countries: Austria, Estonia, France, Germany, Hungary, Ireland, Israel, Italy, Romania, Slovenia and Spain, with Sweden serving as the scientific coordinating centre.

The respective manuscript is a scientifically detailed study protocol that describes each aspect in methodology utilized in the SEYLE project.

Thank you very much.

With Kind Regards,

Danuta Wasserman

Principle Investigator for the SEYLE project